Author's response to reviews

Title: Knowledge, attitudes and practices towards pandemic influenza among cases, close contacts, and healthcare workers in tropical Singapore: a cross-sectional survey

Authors:

Jonathan Yap (jonyap@yahoo.com)
Vernon J Lee (vernonlim@hotmail.com)
Teng Yan Yau (so2_healthcare@starnet.gov.sg)
Tze Pin Ng (Tze_pin_ng@nuhs.edu.sg)
Phern Chern Tor (torphernchern@gmail.com)

Version: 5 Date: 8 July 2010

Author's response to reviews:

Dear Sir/Madam,

I only made 2 minor changes to 2 sentences whereby there was no spacing after the punctuation mark. No changes to content was made.

rgds,
jon